# A comparison of global marine surface-specific humidity datasets from *in situ* observations and atmospheric reanalysis

Elizabeth C. Kent,<sup>a\*</sup>David I. Berry,<sup>a</sup> John Prytherch<sup>a</sup> and J. Brent Roberts<sup>b</sup>

<sup>a</sup> National Oceanography Centre, Southampton, UK

<sup>b</sup> Earth Science Office, NASA Marshall Space Flight Center, Huntsville, AL, USA

**ABSTRACT:** The accurate estimation of near-surface marine-specific humidity is necessary for climate and air-sea interaction applications. Available estimates of monthly mean-specific humidity spanning the past 50-years are based on a variety of sources including *in situ* observations, atmospheric reanalyses and datasets that blend many different data sources. Eight specific humidity datasets are compared and little consensus emerges as to mean values, regional variations and changes over time. For large area averages the datasets do show consistency in their interannual variations and, in the Extratropics, in their seasonal cycles. Adjustments applied to *in situ* observations from ships are shown to be smaller than differences among the datasets and in well-sampled regions and periods the *in situ* data are able to highlight biases in the reanalysis-based specific humidity estimates. Near surface-specific humidity to the assimilation of satellite radiance measurements that became available in 1999. There is less confidence in reanalysis-based estimates of specific humidity over the ocean than over land. However the *in situ*-based humidity analyses have suffered in recent years with a reduction in observation numbers and lack of information on observation methods and heights. Consequently near-surface-specific humidity remains relatively poorly known over the oceans.

KEY WORDS marine climatology; atmospheric reanalysis; surface humidity; air-sea interaction; observations; specific humidity

Received 8 June 2012; Revised 21 November 2012; Accepted 18 February 2013

#### 1. Introduction

Water vapour is the most abundant radiatively active gas in the atmosphere and exerts a strong feedback on the climate system (Trenberth *et al.*, 2007). Nearsurface humidity contrasts drive evaporation from the land and ocean and exert an important influence on the global energy and hydrological cycles. Despite this, surface humidity has historically received less attention than surface temperature, partly because of difficulties associated with humidity measurement. However, recent progress in adjustment and quality assurance of humidity records over both the land and the ocean has resulted in the use of surface humidity records to increase confidence in estimates (Kennedy *et al.*, 2010; Willett *et al.*, 2012) and attribution (Willett *et al.*, 2007) of climate change.

An important motivation for the quantification of marine surface humidity is the estimation of heat and freshwater fluxes between the atmosphere and ocean. Direct measurement of turbulent fluxes is expensive and there are few measurements available (Brunke *et al.*, 2011). Therefore in global marine flux datasets the 'bulk' form of the fluxes is often used where the air-sea

latent heat flux depends on the specific humidity contrast between the sea surface  $(q_s)$  and the air above it (q). The accepted benchmark for surface net heat flux accuracy on large space scales is  $\pm 10 \text{ Wm}^{-2}$  over monthly to seasonal time scales (WCRP, 1989; Webster and Lukas, 1992; WGASF, 2000; Weller *et al.*, 2004; Fairall *et al.*, 2010), implying determination of individual components of the net heat flux to a few Wm<sup>-2</sup>. Where turbulent fluxes are parameterised this sets limit on the mean bias of input specific humidity observations of order 0.3 g kg<sup>-1</sup> (Fairall *et al.*, 1996; Bradley and Fairall, 2007). Other applications, for example global climate change or annual ice mass calculations, require even greater accuracy (Bourassa *et al.*, 2010).

RMetS

Royal Meteorological Society

Humidity of the near-surface atmosphere is designated an Essential Climate Variable (ECV) by the Global Climate Observing System (GCOS, 2010). Observational requirements for climate applications are given in Table 1. Over the ocean, near-surface atmospheric humidity is measured by the Voluntary Observing Ships (VOS) programme and by a small number of fixed platforms (Woodruff *et al.*, 2011). A small number of marine humidity observations date from the 1850s but data are extremely sparse until the 1950s and data coverage peaks between the 1970s and 1980s. Observations from the VOS and its predecessors are made routinely, initially in support of marine climatology (Maury, 1851) and more

<sup>\*</sup> Correspondence to: Dr E. C Kent, National Oceanography Centre, European Way, Southampton, SO14 3ZH, UK. E-mail: eck@noc.ac.uk

Table 1. Observational requirements from the WMO Rolling Review of Requirements. From http://www.wmo-sat.info/db/ requirements (accessed 16 April 2012).

	Goal	Threshold
Uncertainty (g kg <sup>-1</sup> )	1%	2%
Horizontal resolution	25 km	100 km
Temporal resolution	30 min	6 h

recently also in support of weather forecasting (Kent and Ingleby, 2010). Observations are concentrated in the major shipping lanes despite efforts to recruit ships operating in data sparse regions to the programme. Marine surface humidity observations are collated in the International Comprehensive Ocean-atmosphere Data Set (ICOADS, Woodruff et al., 2011) and have been used to produce gridded datasets and analyses of marine surface humidity (Da Silva et al., 1994; Josey et al., 1999; Bourassa et al., 2005; Dai, 2006; Willett et al., 2008; Berry and Kent, 2009). Each of the in situ datasets differs in the version of the source dataset, whether or not buoy observations are included, in the adjustments applied to VOS humidity observations and in the methods used to quality control (QC) and to produce a gridded analysis from scattered observations.

Atmospheric reanalysis model output has become an important resource for understanding the marine atmosphere. Advances in atmospheric modelling, data assimilation techniques and in the observing system have led to sustained improvements in reanalyses over time (Trenberth et al., 2010). Reanalyses provide an internally consistent representation of the atmosphere, but are limited by computational resources, simplifying assumptions in the model physics and can evolve their own climate especially in regions or periods which lack adequate observations to constrain the model (Bosilovich et al., 2011). Because of uncertainty in poorly constrained quantities such as surface fluxes (Rienecker et al., 2011) some researchers have developed blended products, combining reanalysis output with other observational data sources in an attempt to improve the representation of near-surface flux-related variables (Yu and Weller, 2007).

Because in situ measurements of marine surface humidity are sparse in many regions (Berry and Kent, 2009) there have been many attempts to retrieve humidity from satellite observations (Kubota et al., 2002; Bentamy et al., 2003; Chou et al., 2003; Andersson et al., 2010; Roberts et al., 2010; Shie et al., 2010), these products will be considered in another paper. This article will present an evaluation of several of the marine surface-specific humidity fields provided by in situ and reanalysis-based global datasets. Section 2 describes the available in situ observations of humidity from ships and buoys. Section 3 summarizes the characteristics of 10 marine surfacespecific humidity datasets of which 8 are compared in this paper. Section 4 details the processing applied to the datasets to allow this comparison to be made along with adjustments that may be applied for observational height

and measurement bias. Section 5 presents the results of the dataset comparison and some conclusions are drawn in Section 6.

#### 2. In situ measurements of marine surface humidity

## 2.1. Voluntary observing ships

The VOS use a variety of different measurement methods for humidity but metadata indicating observational method is not contained within the ships meteorological reports. Instead, information on humidity measurement methods is obtained from World Meteorological Organisation (WMO) Publication No. 47 (Pub. 47, Kent et al., 2007) and this information has been appended to individual ship observations in the ICOADS (Woodruff et al., 2011). Kent et al. (2007) list the methods that could be entered for the type of humidity sensor and its exposure between 1970 and 2004. The metadata for 2004 was the last year available to Kent et al. (2007) and by this date there was an increase in electric methods of measurement as automated measurements became more common. The most common method is psychrometric: paired wet- and dry-bulb thermometers exposed in either a naturally ventilated screen or a mechanically ventilated psychrometer. Lack of entries for individual VOS in Pub. 47, combined with missing ship identifiers within reports, results in a substantial number of ICOADS humidity reports that do not have any humidity metadata associated with them. This problem has been compounded in recent years by the withholding of ship callsigns from real-time reports in response to security and commercial concerns by the operators of ships participating in the VOS programme. From December 2007 ICOADS contains no callsigns (Woodruff et al., 2011) although some identifier information may become available in the future.

Measurement of humidity at sea requires considerable care and VOS observers are issued with detailed instructions (Met Office, 1995; WMO, 2008). A handbook detailing requirements and methods for research quality observations describes best practices for measuring humidity at sea (Bradley and Fairall, 2007). Observational accuracy of a subset of the VOS providing additional observational metadata in the period 1988-1990 was analysed by Kent et al. (1993a). The VOS humidity data were classified according to whether they were measured using a mechanically ventilated instrument (denoted psychrometer) or by a naturally ventilated instrument (denoted screen). It concluded that psychrometer-measured humidities were on average drier than those measured using screens. There was some evidence that humidities measured by an Ocean Weather Ship (OWS) were drier still, but regional effects on the comparison could not be ruled out. Although VOS air temperature observations contain significant diurnal biases due to heating of the ship environment (Kent et al., 1993b; Berry et al., 2004; Berry and Kent, 2005), this is not expected to cause comparable biases in measurements of dew point temperature or specific humidity (Kent *et al.*, 1993a; Kent and Taylor, 1996). It should be noted, however, that relative humidity will be underestimated if calculated using the erroneously high air temperature (Dai, 2006). A further potential source of bias occurs if an adjustment to a standard reference height is applied using estimates of atmospheric stability based on air temperature observations containing significant diurnal biases. The methods to measure humidity are changing over time and there has not yet been an assessment of the electric sensors that have recently come into more common use (Kent *et al.*, 2007).

Measurements require adjustment to a standard reference height as in the surface layer the water vapour content decreases with distance from the sea surface. Adjustment from the measurement height to the marine standard reference height of 10 m is typically a few tenths  $g kg^{-1}$ . Monin-Obukhov similarity theory (MOST; Smith, 1980, 1988) is typically used to make the adjustment and requires an estimate of the stability of the atmospheric surface layer. Stability is estimated from the near-surface wind speed, air temperature, humidity and the sea surface temperature (SST). Measurement heights for the VOS are not contained within the ship report but are available from Pub. 47. Although the height of the humidity sensor is not reported it can be estimated from available information: observing platform height (from 1968); barometer height (from 1995) and air temperature sensor height (from 2002), see Kent et al. (2007) for further information. Because of the lack of ship identifiers measurement heights have not been associated with individual ICOADS observations from December 2007.

#### 2.2. Moored buoys

Because of the need to replenish reservoirs for wetbulb measurements, autonomous humidity measurements from moored buoys have been a relatively recent development. Challenges include power limitations, stability in the presence of contaminants and durability (Weller et al., 2008). Trials of humidity sensors on operational moored buoys in the late 1980s revealed problems with reliability (Breaker et al., 1998). Sensors were added to the Tropical Atmosphere Ocean moored buoy array in 1989 and were assessed as having root-mean-square errors in relative humidity of 4% (McPhaden et al., 1998). Improved sensors were deployed in 1993 on 2 buoys in the Gulf of Mexico and were shown to meet WMO standards for reliability (Breaker et al., 1998) and humidity is now measured on many moored buoys (McPhaden et al., 1998, 2009; Bourlès et al., 2008). Buoy observations are typically made at heights of about 3 m for the Tropical arrays but many Extratropical buoys are larger and therefore measurement heights may be greater outside the Tropics. For the Tropical moored buoy arrays, measurements from the Pacific are within  $10^{\circ}$  of the Equator, measurements from the Atlantic and Indian Oceans are within  $20^{\circ}$  of the Equator. Extratropical buoys are usually coastal. Measurements of humidity from buoys were

excluded from the analysis of Berry and Kent (2009) but included by other analyses (Da Silva *et al.*, 1994; Josey *et al.*, 1999; Bourassa *et al.*, 2005; Willett *et al.*, 2008).

## 3. Marine surface humidity datasets

#### 3.1. In situ datasets

## *3.1.1. UWM/COADS*

The longest in situ humidity dataset was produced by the University of Wisconsin-Milwaukee (UWM) based on the Comprehensive Ocean-atmosphere Data Set (COADS, the predecessor to ICOADS) Release 1 (Woodruff et al., 1987). The UWM/COADS dataset (Da Silva et al., 1994) covers the period 1945-1989 and presents an analysis of monthly means on a  $1^{\circ}$  grid for the global ocean. It is noted that only features with scale greater than 770 km (1300 km below  $40^{\circ}$ S) are retained by the analysis (Da Silva et al., 1994). The observations are interpolated to a regular grid using successive correction of  $1^{\circ}$  monthly box averages (following Cressman, 1959) that does not account for the varying uncertainty of the box averages (Kent et al., 2000). Specific humidity is adjusted to a 10 m reference height  $(q_{10})$  from an assumed measurement height of 20 m (A. M. da Silva, pers. comm. 2012). No adjustments are applied to the humidity data for observational bias. QC is applied using COADS 'trimming flags' to remove data outside of 3.5 standard deviations of a climatological median. No ice mask is applied but in areas of climatological ice cover (Alexander and Mobley, 1976) a tighter QC limit of 2.8 standard deviations is applied.

## 3.1.2. NOC Surface Flux dataset v1.1

NOCv1.1 (previously called the SOC Air-Sea Flux Climatology) is based on COADS Release 1a (Woodruff et al., 1993). The analysis uses successive correction but with a minimum radius of 331 km (c.f. 770 km used in UWM/COADS) and presents monthly means on a  $1^{\circ}$  grid for the period 1980-2005. Kent et al. (2000) estimated the spatial resolution of the NOCv1.1 dataset to have a mean resolution of about  $4^{\circ}$  latitude by  $6^{\circ}$  longitude with a modal value of  $3^{\circ}$  in each direction. QC uses the ICOADS trimming limits to exclude data outside of 4.5 standard deviations from a long-term median value. Adjustments to the humidity observations from screens (defined following Kent et al., 1993) are made by applying a linear correction to the dew point temperatures (Josey *et al.*, 1999) and adjusted to give  $q_{10}$ . Height and measurement method metadata from Pub. 47 are used, where available. In the absence of metadata, it is assumed that the measurement height for ships is 18 m and that one third of the observations is made using naturally ventilated screens. The data, methods and adjustments applied in the construction of NOCv1.1 are summarised and compared to the methods used in NOCv2.0 by Berry and Kent (2011).

# 3.1.3. FSU v3.0

The methods used to generate the Florida State University (FSU) Flux Dataset version 3.0 are described by Bourassa et al. (2005) and Smith et al. (2011). The input observations are from ICOADS Release 2.2 (a minor update from Release 2.1; Worley et al., 2005). FSU v3.0 is available from 1978 to 2004 on a monthly  $1^{\circ}$ grid; we use the version that is unfiltered in time (as used by Smith et al., 2011; S. R. Smith, pers. comm. 2012). QC is applied based on 3.5 standard deviations difference from the UWM/COADS dataset (Da Silva et al., 1994). In regions where the standard deviation in the UWM/COADS datasets is limited by poor sampling, minimum standard deviation limits are prescribed (Bourassa et al., 2005). A variational method using multiple constraints with objective weights is used to generate the fields (Bourassa et al., 2005). Specific humidity is adjusted to give  $q_{10}$  assuming a measurement height of 20 m and no other adjustments are applied.

## 3.1.4. Dai

This combined land and ocean dataset covers the period December 1975 to May 2005 and was based on ICOADS Release 2.1 (Worley et al., 2005) plus ICOADS near realtime updates from 2002. Dew point, air temperature and pressure observations are range checked and those observations outside 4.5 standard deviations of a  $1^{\circ}$  grid box mean for the entire period excluded. Specific humidity and relative humidity are calculated and subject to similar range and standard deviation checks. Those observations passing QC are averaged in  $4^\circ$   $\times$   $5^\circ$  latitude/longitude seasonal grid boxes (Dai, 2006). Adjustments are not applied for measurement height or for observational bias. However, relative humidity is calculated using air temperatures adjusted for solar heating biases following Kent et al. (1993) noting that the specific humidity was calculated using unadjusted air temperatures (Kent and Taylor, 1996). Because not all the information required to adjust the air temperature is available for the entire period of analysis, unadjusted air temperatures are used to calculate relative humidity anomalies. Dai (2006) notes that although the adjustments to air temperature change the diurnal cycle and mean values of air temperature and relative humidity, the impacts on the interannual and longer-term variability are small. Analysis of the resulting dataset indicated a statistically significant reduction over time in relative humidity over the ocean. The analysis also indicated statistically significant upward trends in specific humidity over much of the Atlantic and parts of Indian and Pacific Oceans.

# 3.1.5. HadCRUH

The HadCRUH dataset is a combined land and ocean specific and relative humidity anomaly only dataset presented on a  $5^{\circ}$  monthly grid covering the period 1973–2003. The presentation of anomalies avoids the additional uncertainty inherent in the calculation of absolute values (Dai, 2006). Over the oceans, HadCRUH is

based on ICOADS Release 2.1 (Worley et al., 2005) supplemented with ICOADS near real-time updates from 1998 onward. No adjustments are made for either height or observational bias. Specific humidity and relative humidity are calculated using unadjusted dew point and air temperatures. Heating biases in air temperatures should not affect the calculation of specific humidity (Kent and Taylor, 1996) but will result in an underestimate of the relative humidity. However, this should have a smaller effect on the relative humidity anomalies presented in HadCRUH than on the mean values (Dai, 2006). QC is applied to the marine data using differences from climatology, outlier and neighbour consistency checks, iterating to produce new climatology estimates created using only data passed by the previous round of QC until only a small number of observations are removed (Willett et al., 2008). Gridding methodology follows Rayner et al. (2006). In mixed ocean-land  $5^{\circ}$  grid boxes the land and marine estimates are averaged weighted according to the proportional presence of land and ocean in the box, but with a minimum contribution from each of 25% (Willett et al., 2007). There is no indicator in the dataset to differentiate between land, ocean and mixed grid boxes.

Analysis of the marine anomalies (Willett *et al.*, 2007) showed an increase over time in specific humidity and a decrease in relative humidity statistically significant at the 1% level for global data. It was suggested that the decrease in relative humidity may have been at least partly due to a moist bias, apparent in the humidity data prior to 1982 and possibly related to changes in observing practice, but this could not be confirmed (Willett *et al.*, 2008). Willett *et al.* (2012) showed that for the period 1976–2003 the variability and trends in marine  $q_{10}$  calculated from HadCRUH and Dai (2006) are similar.

## 3.1.6. NOC Surface Flux dataset v2.0

NOCv2.0 is based on ICOADS Release 2.4 (Worley et al., 2005) up to and including 2006 and ICOADS Release 2.5 (Woodruff et al., 2011) thereafter including preliminary products from 2008. The dataset as analysed here covers 1971-2010. Only ship observations are used. Adjustments are applied to give  $q_{10}$  using known heights where available (Kent et al., 2007) or default values based on a  $2^{\circ}$  area monthly gridded dataset of humidity measurement heights (Berry and Kent, 2009). Bias adjustments are applied to screen-measured humidities by reducing the specific humidity estimates by 3.4% (Berry and Kent, 2011). Where no measurement method was available a partial adjustment was made based on the fraction of observations with known method within the same month and  $10^{\circ}$  area made using screens (Berry and Kent, 2009). Due to the loss of callsign information in ICOADS, from 2007 onwards the last calendar month with information was used to estimate the fraction of screens in use and the fractional correction to apply, i.e. the data for January 2006 were used for all subsequent Januarys etc. The dataset is constructed using Optimal Interpolation (OI) with a Gaussian space scale of 300 km and a time scale of 3 d to produce daily fields on a 1° grid. The random uncertainty used to weight  $q_{10}$  relative to the background field of  $q_{10}$  is 1.3 g kg<sup>-1</sup> calculated following Kent and Berry (2005). Uncertainty estimates are produced for each day and grid box and are combined to give monthly mean values and uncertainties accounting for the correlation in errors between days (Berry and Kent, 2011).

Trends in  $q_{10}$  from NOCv2.0 are similar to those shown by Dai (2006) and in HadCRUH over the period 1976–2003 (Willett *et al.*, 2012). Differences are due to the adjustments applied to NOCv2.0 and are particularly noticeable prior to about 1985.

## 3.2. Reanalyses based datasets

## 3.2.1. 20CRv2

The 20th Century Reanalysis version 2 (20CRv2; Compo et al., 2011) provides 6-hourly, daily and monthly fields of  $q_2$  on a 2° grid from 1871 to 2010. 20CRv2 is a global atmospheric circulation dataset, assimilating only surface pressure reports over both the land and ocean and using observed monthly sea-surface temperature and sea-ice as boundary conditions. It uses an Ensemble Kalman Filter data assimilation method with background 'first guess' fields supplied by an ensemble of forecasts from a global numerical weather prediction model. This directly yields a global analysis every 6h as the most likely state of the atmosphere, and also an uncertainty estimate of that analysis through the spread of the ensemble. The SST and sea-ice come from the HadISST dataset (Rayner et al., 2003) and marine pressure observations come from ICOADS Release 2.4 (from 1952) and ICOADS Release 2.5 (from 1871). No humidity observations are assimilated into 20CRv2.

## 3.2.2. ERAI

The European Centre for Medium-Range Weather Forecasts (ECMWF) Interim Reanalysis (ERAI) covers the period 1979 to near present (November 2011 at the time of writing) and the version analysed here is available on a 6-hourly  $1.5^{\circ}$  grid. There were problems with the representation of the hydrological cycle in the previous ECMWF reanalysis (ERA40, Uppala et al., 2005) and improvements have been developed (Andersson et al., 2005) and implemented in ERAI (Dee et al., 2011). In ERAI a four-dimensional analysis provides fields based on forecast model output with data assimilation (ECMWF, 2007), followed by a separate OI analysis of near-surface parameters, including  $q_2$  and 2 m temperature (T2; Dee et al., 2011). Several different SST and sea-ice concentration datasets are used with changes in July 2001, January 2002 and February 2009 (Dee et al., 2011).

Many different observational sources are assimilated into the forecast model but over the oceans, observations from the microwave sounding instruments [Special Scanning Microwave/Imager (SSM/I) and the Advanced Microwave Sounding Unit-B (AMSU-B)] dominate the humidity analysis (Andersson *et al.*, 2007). The 2 m OI analysis uses background values derived from the lowest model level using MOST. Surface observations of marine humidity from ships and buoys are used in the 2 m OI analysis without any adjustments (A. J. Simmons, pers. comm. 2008; Simmons *et al.*, 2010). Doubts about the quality of input observations of temperature and humidity from ships meant that background rather than assimilated fields were used in a recent climate assessment (Willett *et al.*, 2012). The water cycle in ERA-Interim is thought to be too intense over the oceans (Dee and National Center for Atmospheric Research Staff, 2012).

Temperature and specific humidity are prognostic variables (ECMWF, 2007) but 2 m dew point temperature  $(T_{d2})$  is archived rather than  $q_2$ .  $T_{d2}$  is computed from the estimate of  $q_2$  and surface pressure.  $T_{d2}$  uses the saturation formulation with respect to water: if the resulting  $T_{d2}$  is higher than the  $T_2$ ,  $T_{d2}$  is set equal to  $T_2$  (ECMWF, 2007; D. P. Dee., pers. comm. 2012).

## 3.2.3. MERRA

The National Aeronautics and Space Administration Modern Era Retrospective-analysis for Research and Application (MERRA) had a particular focus on improving the representation of the hydrological cycle compared to previous reanalyses (Rienecker et al., 2011). MERRA data is available from 1979 to present both as  $0.667^{\circ}$  ×  $0.5^{\circ}$  6-hourly fields and monthly mean fields. The version analysed here is the monthly dataset. MERRA makes values of both  $q_2$  and  $q_{10}$  available. MERRA assimilates many different observation types (Rienecker et al. 2008, 2011) including in situ humidity from radiosondes, ships and buoys and satellite radiances from several different satellite sources (Rienecker et al., 2008) of which the most important for humidity estimation are SSM/I and AMSU-A (Robertson et al., 2011). MERRA is forced at the sea surface by SST and sea-ice concentrations from the weekly  $1^{\circ}$  analysis of Reynolds *et al.* (2002) interpolated to each model time step (Rienecker et al., 2011). Atmospheric temperature and humidity at the lowest model level are prognostic variables and are affected by the boundary layer parameterisation and by assimilated observations (Roberts et al., 2012). The thickness of the lowest model level is variable with a mean value over the ice-free ocean of approximately 130 m, varying from about 120 m at high latitudes to about 134 m near the Equator. Humidity at the 2 and 10 m levels is calculated as a diagnostic variable using similarity theory with values at the surface and the lowest model level. It should be noted that at low wind speeds and especially in stable conditions, estimates at these heights may be decoupled from surface values (Smith, 1988).

The water budget in MERRA has been studied by Bosilovich *et al.* (2011) who note the impact of the assimilation of data from AMSU starting in November 1998.

Analysis of MERRA output before and during the assimilation of AMSU data shows substantial changes in mean values, annual cycles and phase of variables related to the hydrological cycle. Assimilation of AMSU-A data leads to greater total column water over the ocean and an intensification of the hydrological cycle. The analysis increments, which indicate the changes that observations make to the model forecast, show the largest changes are in tropical and high latitude regions. The assimilation of observations is performed to correct biases in the estimated atmospheric state due to deficiencies in the model physics. The analysis increments have complex dependencies on the ability of the model to represent the processes that define the atmospheric state at any particular place and time and also on the availability and accuracy of observations that may constrain the model. Analysis increments have been used by Robertson et al. (2011) to examine the impact of observations on water fluxes in MERRA. For zonally averaged (land and ocean) analysis increments of vertically integrated moisture, the impact of first SSM/I in August 1987 and then AMSU-A in November 1998 is clear (Robertson et al., 2011). It should be noted that similar discontinuities, particularly in evaporation, are also present in other reanalyses such as the National Centers for Environmental Prediction (NCEP) Climate Forecast System Reanalysis (CFSR, Wang et al., 2010).

#### 3.3. Blended datasets

## 3.3.1. OAFlux

The Objectively Analyzed air-sea Fluxes for the Global Oceans (OAFlux) is a blended product combining humidity fields from reanalyses and satellites (Yu et al., 2008) covering the period from 1958 to mid-2011. The National Centres for Environmental Prediction Reanalysis version 1 (NCEP1, Kalnay et al., 1996) is used throughout the analysis. NCEP version 2 (NCEP2; Kanamitsu et al., 2002) is used from its start in 1979. The ECMWF 40year Reanalysis (ERA-40; Uppala et al., 2005) is used from 1958 to mid-2002. The satellite surface humidity fields constructed by Chou et al. (2003) from the SSM/I series of sensors were used from mid-1987 to the end of 2000. The three reanalysis models provide  $q_2$  whilst the SSM/I estimates are representative of  $q_{10}$ . The SSM/I estimates have been adjusted to a 2-m reference height, i.e.  $q_2$ , using the COARE3 algorithm (Fairall et al., 2003) and the near-surface humidity from OAFlux is representative of  $q_2$  (Yu and Weller, 2007). Weights for each field are calculated objectively using researchquality buoy observations (Yu et al., 2008) and are: NCEP1 = 1; NCEP2 = 1; ERA-40 = 1.3; SSM/I = 0.4. It should be noted that many different types of observation at the surface and through the atmosphere are assimilated into the reanalyses in differing combinations and with different data availability, QC and weighting. It is therefore difficult to determine the dependency of the OAFlux humidity estimates on any particular observation type.

#### 4. Data processing

#### 4.1. Regridding and masking

Each of the available datasets has been regridded onto the same  $1^{\circ}$  latitudinal/longitudinal monthly grid (Table 2 and 3). NOCv1.1a and Dai (2006) are not analysed in this paper. NOCv2.0 covers a longer period than NOCv1.1a and the methods used to calculate humidity fields have now been updated. The Dai (2006) dataset has not been made publically available. UWM/COADS, FSUv2.0, NOCv2.0 and OAFlux are already available on this grid (Table 2) and are ocean-only so no further processing is required. Ice free, ocean only grid boxes for MERRA, ERAI and 20CRv2 were selected using the land and ice masks supplied as part of those datasets. Ocean only grid boxes were selected from HadCRUH using the ERAI land mask. HadCRUH and 20CRv2 are available on 5° and 2° grids respectively and the 1° values have been assigned as the value from the nearest  $5^\circ$  or  $2^\circ$  grid box.  $q_2$  is not available for ERAI and has been calculated from the archived 6-hourly, 1.5° latitude/longitude values of T<sub>d2</sub>, T<sub>2</sub> and surface pressure following Henderson-Sellers (1984). The  $q_2$  estimates have then been averaged to give estimates of monthly mean  $q_2$ . For a limited period estimates of  $q_{10}$  have been calculated using surface pressure,  $T_{d2}$ ,  $T_2$ , 10-m wind speed and SST data. The monthly  $1.5^{\circ}$  grid box values are then linearly interpolated to the monthly  $1^{\circ}$  grid. MERRA  $q_2$  and  $q_{10}$  values for "open water" grid boxes were linearly interpolated from the  $0.667^{\circ} \times 0.5^{\circ}$  grid onto the  $1^{\circ}$  grid.

Figure 1 shows masks that have been applied to the datasets to account for the presence of land, sea-ice and regions of sparse data availability. NOCv2.0 and UWM/COADS are both masked according to the estimates of total uncertainty in NOCv2.0. 1° grid boxes where the average  $q_{10}$  uncertainty is greater than  $1\,\mathrm{g\,kg^{-1}}$  or the air temperature uncertainty is greater than 1 °C are excluded (Figure 1(a)). FSUv3.0 does not provide estimates in regions with particularly poor sampling (Figure 1(b)). HadCRUH has been masked to only include  $5^{\circ}$  grid boxes where it contains data for every month (Figure 1(c)). Where only reanalysisbased datasets are compared a common mask is applied based on the sea-ice concentration fields from NOCv2.0, which are themselves based on Reynolds et al. (2002; Figure 1(d)). The mask is invariant over time and excludes data where the sea-ice concentration is above zero in any month during the analysis period. Where in situ datasets are compared a mask combining the NOCv2.0 and the FSUv3.0 masks is applied (Figure 1d)

## 4.2. Adjustments to marine humidity observations

Some datasets have adjustments applied to account for expected differences due to measurement method and height (Table 2). Adjustment is problematic because the required metadata on observation heights and methods is not always available and there is little high-quality validation data for humidity over the ocean (Roberts *et al.*, 2012).

			T	aute 2. Dataset characteristic	o.			
Dataset	Input data	<i>In situ</i> humidity platforms	Period	Grid	Reference height (m)	Height adjustment	Bias adjustment	Coverage
UWM/COADS	COADS R1 &	Ships Buoys	1945-1989	$1^{\circ}$ monthly	10	From 20 m	None	Ocean
NOCv1.1a	COADS R1a	Ships Buoys	1980 - 2005	$1^{\circ}$ monthly	10	Based on Pub. 47	Yes (Josey	Ocean (with
Dai	Plus updates ICOADS R2.1	Ships	December 1975	$4^{\circ} \times 5^{\circ}$ seasonal	10	None	et at., 1999) None	Global
HadCRUH	plus updates ICOADS R2.1	Ships Buoys	1973–2003	$5^{\circ}$ monthly	anomaly	None	None	Global
FSUv3.0	plus updates ICOADS R2.2	Ships Buoys	1978-2004	$1^{\circ}$ monthly	10	From 20 m	None	Ocean (with
NOCv2.0	ICOADS R2.4	Ships	1971-2010	$1^{\circ}$ monthly	10	Based on Pub. 47	Yes (Berry and	Ocean
20CRv2	COADS R2.4	None	1871-2010	$2^{\circ}$ monthly	2	N/A	N/A	Global
ERAI	See Dee et al.	Ships Buoys	1979-2011	$1.5^{\circ}$	7	None	None	Global
MERRA	See Rienecker	Ships Buoys	1979-2011	$0.667^{\circ} \times 0.5^{\circ}$ monthly	2, 10	None	None	Global
OAFlux	ce u. (2000) See Yu <i>et al.</i> (2008)	Ships Buoys	January 1958 to July 2011	$1^{\circ}$ monthly	7	None	None	Ocean

Table 2 Dataset characteristics



Figure 1. Masks applied to the datasets: (a) NOCv2.0 and UWM/COADS; (b) FSUv3.0; (c) HadCRUH; (d) reanalysis sea-ice and land mask (dark grey as used for 20CRv2, ERAI, MERRA and OAFlux), and combined *in situ* mask (light grey as used for UWM/COADS, NOCv2.0 and FSUv3.0).

Table 3. Dataset availability. All were accessed on 14th November 2012.

Dataset	Website
UWM/COADS	http://iridl.ldeo.columbia.edu/SOURCES/.DASILVA/.SMD94/
NOCv1.1a	http://www.noc.soton.ac.uk/noc_flux/noc1_1.php
Dai	Not publically available
HadCRUH	http://www.metoffice.gov.uk/hadobs/index.html
FSUv3.0	http://coaps.fsu.edu/RVSMDC/FSUFluxes/index.php
NOCv2.0	http://rda.ucar.edu/datasets/ds260.3/
20CRv2	http://www.esrl.noaa.gov/psd/data/gridded/data.20thC_ReanV2.html
ERAI	http://data-portal.ecmwf.int/data/d/interim_full_moda/
MERRA	http://disc.sci.gsfc.nasa.gov/daac-bin/DataHoldings.pl
OAFlux	http://oaflux.whoi.edu

Comparison of humidity datasets is also complicated by the difference in reference heights chosen, the range of different humidity variables provided and the nonlinearity of the relationships among humidity, temperature and pressure. Over the ocean the reference height used to adjust observations is typically 10 m, for land observations it is typically 2 m ("screen height").

The effects of adjustments to VOS-specific humidity measurements are examined using the NOCv2.0 dataset (Berry and Kent, 2009, 2011). Three versions of the dataset have been constructed, one with no adjustments applied, one with only the height adjustment applied and one with the full adjustment for both height and measurement method. The height adjustment of humidity data increases the specific humidity estimates on average. This is because most of the measurements are made above the marine reference height of 10 m (Kent *et al.*, 2007) and specific humidity typically decreases with height in the surface layer. In contrast the adjustments made for measurement method act to decrease the estimated specific humidity.

Both adjustments depend on the value of specific humidity itself, so are larger in warmer regions where the specific humidity is high. The adjustments also reflect regional and temporal differences in measurement height and methods (Kent *et al.*, 2007). The availability of height and measurement method metadata varies greatly over time. This is sometimes because the metadata itself is missing, but is often because metadata cannot be associated with individual ICOADS reports because ship identifiers are missing. For most of the period after 1971 the availability of metadata improves over time, but after 2007 ICOADS callsigns and hence humidity method metadata are missing. Where metadata cannot be associated with individual reports adjustments are made based on available information. This is inherently uncertain and relies on the assumptions that the proportions of observations from the different methods are the same for the known and unknown cases. Similarly, the unknown heights are assumed to be similar to the known heights. The observing system characteristics are assumed to vary slowly over time. Recently Kennedy et al. (2011) approached the problem of unknown metadata by generating an ensemble of equally likely realisations of SST fields to account for uncertainty in metadata and adjustments. A similar approach would help to improve our understanding of uncertainties in bias-adjusted humidity fields but has not yet been attempted.

Figure 2 shows time series of the specific humidity adjustments based on NOCv2.0 for the period 1971-2006 for which metadata are available. The adjustment for bias in the screen-measured-specific humidities scales with humidity so is largest in regions and seasons of higher humidity, but modulated by the relative proportions of screens and psychrometers. Figure 3 shows averages over the period 1971-2006 of the adjustments. Annual averages of the height adjustment applied to the NOCv2.0 dataset over the masked region shown in Figure 1(a) increase by  $0.11\,g\,kg^{-1}$  between 1971 and 2006 (the specific humidity is increased in 1973 by  $\sim 0.08 \,\mathrm{g \, kg^{-1}}$ and in 2006 by  $\sim 0.19 \text{ g kg}^{-1}$ ). This picture is complicated by the adjustments applied for measurement method that are larger (on average the humidity is dried by  $\sim 0.26 \,\mathrm{g \, kg^{-1}}$ ) and have a more complicated spatial and temporal structure (Figures 2 and 3). The most important effect of the bias adjustments is an overall decrease of the specific humidity with a strong regional dependence in the size of the decrease. It should be noted that if no height correction is applied or the heights are assumed constant over time a small artificial drying trend would be present in the data due to the increase in observing height with time.

It should also be noted that the adjustments applied to NOCv2.0 are not well-represented by a linear trend and variations in the bias adjustment are small compared with both the annual cycle and the interannual variability.

## 4.3. Surface humidities in MERRA and ERAI

For the period 1989–2009 monthly mean  $q_{10}$  was calculated using 6-hourly values from ERAI using MOST and the transfer coefficients of Smith (1980, 1988). MERRA provides values for both  $q_2$  and  $q_{10}$  and differences between these values ( $\Delta q_{2-10}$ ) are shown in Figure 4 along with  $\Delta q_{2-10}$  calculated for ERAI. Mean  $\Delta q_{2-10}$  has a strong zonal dependence and is about 50%

larger for MERRA than for ERAI (Figure 4). Roberts et al. (2012) showed that compared with research vessel observations the specific humidity difference in MERRA between the sea surface and 10 m ( $\Delta q_{s-10}$ , denoted QSQA by Roberts et al., 2012) was overestimated at low specific humidity values (e.g. by  $1 \text{ g kg}^{-1}$  for specific humidities below  $3 g kg^{-1}$ ) and underestimated at higher specific humidity values (e.g. by nearly  $1.5 \, g \, kg^{-1}$  for specific humidities above 9 g kg<sup>-1</sup>). The values of  $\Delta q_{s-10}$ as calculated for ERAI are on average larger than those for MERRA in each of the ranges of specific humidities quoted by Roberts et al. (2012). There is a small decrease in  $\Delta q_{2-10}$  in the latter part of the MERRA analysis which is particularly noticeable in the tropics (not shown) and may be related to the assimilation of radiances from the AMSU-A starting in late 1998 (Bosilovich et al., 2011; Robertson et al., 2011; Roberts et al., 2012). In contrast, the ERAI values of  $\Delta q_{2-10}$  calculated using MOST increase slightly over the period 1989–2009.

Figure 5(a) shows global average differences between MERRA and ERAI values of  $q_s$ ,  $q_2$  and  $q_{10}$ .  $q_s$  has been calculated from MERRA SST by Roberts et al. (2012) for the period 1979-2007 and for ERAI in this study for the period 1989-2010. The global mean difference between in  $q_s$  in MERRA and ERAI increases by slightly less than  $0.1 \,\mathrm{g \, kg^{-1}}$  at the start of 2002 which coincides with a change in the SST and sea-ice concentration forcing fields used by ERAI (Dee et al., 2011). Global mean  $q_2$  and  $q_{10}$  both increase in MERRA relative to ERAI over the period 1989-2007 (see also Figure 4(c) and (f)). Figure 5(b) shows the zonal average difference between  $q_{\rm s}$  in MERRA and ERAI, smoothed with a 12-month moving average filter. ERAI  $q_s$  is typically higher than that from MERRA near the equator and at high latitudes and MERRA  $q_s$  higher elsewhere. The change in the ERAI SST forcing in 2002 is clear from Figure 5(b), after 2002 zonal annual mean  $q_s$  in MERRA is higher than in ERAI at most latitudes. There is an increase from 2001 in the annual cycle of the differences between MERRA and ERAI in both  $q_2$  and  $q_{10}$  for which the reasons are unclear.

Figure 6 shows MERRA ocean-only analysis increments for vertically integrated water vapour (liquid and ice water increments are much smaller) averaged in three periods: prior to the launch of SSM/I (period 1); SSM/I available but prior to the launch of AMSU-A (period 2) and after the launch of AMSU-A (period 3). Also shown are zonal averages of the ocean-only increments with a 12-month moving average filter applied. We note that these are total column water vapour amounts not near-surface humidity and that transport processes may lead to changes that are non-local, including over land (Bosilovich et al., 2011 note increased ocean to land transport of water after 1999). In period 1, the analysis increments are negative (i.e. drying) at high latitudes and in the tropics and positive (i.e. wetting) in mid-latitudes (more strongly in the Northern Hemisphere where the *in* situ observing system is concentrated). Localised regions of positive moisture increments are seen, some of which



Figure 2. Adjustments applied to NOCv2.0 dataset over the period 1971–2006 (g kg<sup>-1</sup>). Annual mean (left panels); seasonal cycle relative to annual mean (centre panels); mask showing data averaged (right panels). Solid line is the height adjustment, dashed line is the bias adjustment and the dotted line is the combined height and bias adjustment. (a) 25°N–75°N; (b) 25°S–25°N; (c) 75°S–25°S; (d) 75°S–75°N.

are probably associated with radiosonde observations on Pacific Islands. Period 2 (SSM/I) shows an increase of the negative moisture increments in high latitudes and of positive increments in mid-latitudes. In the tropics, there are strong regional variations that give an overall reduction in drying increments over period 2. Period 3 gives the most dramatic change (Bosilovich *et al.*, 2011; Robertson *et al.*, 2011; Roberts *et al.*, 2012) with widespread positive increments and only localised regions where the increments remain negative. The impact of the changing observing system on the integrated water vapour is clear from a Hovmöller diagram of analysis increments smoothed with a 12-month moving average filter (Figure 6(d)).

## 5. Results

5.1. Reanalysis-based products, ocean/ice masking only applied

Figure 6 shows annual average values and seasonal cycles of q for the reanalysis products in three latitude bands. In the Northern region (25°N to 75°N, Figure 7(a)) the interannual variability is similar between all the products, but the mean values show substantial differences. All the products (ERAI, MERRA, 20CRv2 and OAFlux) have a reference level of 2 m, MERRA  $q_{10}$  is also plotted. MERRA  $q_2$  is highest averaged over the region, the  $q_{10}$ values from MERRA agree better with the  $q_2$  values from the other datasets. The other datasets show similar area



Figure 3. Adjustments applied to NOCv2.0 dataset over the period 1971-2006 (g kg<sup>-1</sup>). (a) Height adjustment; (b) bias adjustment; (c) combined height and bias adjustment; (d) zonal mean adjustments averaged over 1971-1975. Solid line is the height adjustment, dashed line is the bias adjustment and the dotted line is the combined height and bias adjustment; (e) as (d) but for 2001-2005.

annual average  $q_2$ , agreeing to within about  $0.1 \text{ g kg}^{-1}$ until the mid-1990s. After this time the  $q_2$  estimates diverge: ERAI shows the smallest increase in humidity and OAFlux a slightly larger increase than either 20CRv2 or MERRA. By 2010, the gap between ERAI and OAFlux averaged humidities has increased to about  $0.3 \text{ g kg}^{-1}$ . The seasonal cycle in the Northern Extratropics is large, about  $6 \text{ g kg}^{-1}$  and there is good agreement amongst the different products.

In the Tropics ( $25^{\circ}$ S to  $25^{\circ}$ N, Figure 7(b))  $q_2$  is higher and differences among products larger compared with those to the North although interannual variations remain similar. OAFlux and MERRA  $q_2$  are now much more similar although MERRA  $q_2$  is increasing more rapidly than OAFlux. q2 from ERAI and 20CRv2 remains more similar to  $q_{10}$  from MERRA. Changes over time in OAFlux and 20CRv2 remain similar, both showing an increase over the entire period. Since the 1980s,  $q_2$  in ERAI remains fairly constant (representing a drying in relative humidity as temperature has increased) and MERRA shows the strongest increase among these datasets in the Tropics. The annual cycle is much smaller than in the Northern Extratropics at less than  $1 \text{ g kg}^{-1}$ . ERAI shows the strongest annual cycle in the Tropics and 20CRv2 the weakest. 20CRv2 shows peak humidities a month later than the other datasets and the minimum humidity seen in the other products in August is not seen.

In the Southern Extratropics  $(75^{\circ}S \text{ to } 25^{\circ}S, Figure 7(c))$ , there is less correspondence between the interannual variability among the reanalysis products, probably as they are less constrained by the *in situ* observing system in this region. As in the Northern Extratropical region ERAI  $q_2$  is driest of all the reanalysis-based products and MERRA shows the highest  $q_2$ . OAFlux shows the strongest increase in  $q_2$  over time. There is fairly good agreement in the reanalysis-based products.

Annual averages for the global ice-free ocean are shown in Figure 7(d). 20CRv2 and OAFlux show similar magnitudes and change over time with OAFlux showing a slightly larger increase. Globally MERRA  $q_2$  starts highest and shows the strongest increase over time. ERAI shows the lowest  $q_2$  of any of the products in the global average (only MERRA  $q_{10}$  is lower) and also shows very little change over time. The difference between MERRA and ERAI  $q_2$  increases from 0.4 to 0.8 g kg<sup>-1</sup> over the period 1979–2010. The global seasonal cycle is small ~0.4 g kg<sup>-1</sup>.

Figure 8 shows the differences between decadal zonal averages of  $q_2$  for the reanalysis-based products using OAFlux as a reference. Only OAFlux and 20CRv2 are available over the full period from 1960 and 20CRv2 is not suitable for use as a reference as in these decadal averages there is a signature of the spectral nature of the



Figure 4. MERRA and ERAI  $q_2-q_{10}$  (g kg<sup>-1</sup>) 1990–1999. (a) MERRA average; (b) MERRA January (solid) and July (dotted) zonal averages; MERRA monthly mean over ice-free ocean; (d-f) as (a-c) but for ERAI.



Figure 5. (a) monthly mean MERRA-ERAI specific humidity (g kg<sup>-1</sup>) averaged over area shown in Figure 1(d): dotted =  $q_{10}$ ; solid =  $q_2$  and dashed =  $q_s$ . (b) MERRA-ERAI zonal mean  $q_s$  over same region as (a) and with 12-point moving average filter. Solid line is zero contour.

20CRv2 model evident as oscillations in long-term mean values (G. P. Compo, pers. comm. 2010). A 5° latitude moving average filter has therefore been applied to the zonal average of the 20CRv2 data plotted in Figure 8 to reduce the visual impact of the fluctuations. Filtered and unfiltered values are shown in Figure 8(b), filtered values only in Figures 8(c-f). It should be noted that OAFlux

is not expected to be homogeneous over the analysis period as the input reanalyses show inhomogeneity over their period of record (Sterl, 2004) and the availability of different input products varies over the OAFlux period of record (Yu *et al.*, 2008). We also note that OAFlux shows the strongest increase in  $q_2$  of any of the products in the Extratropics.



Figure 6. MERRA analysis increments for integrated water vapour (kg m<sup>-2</sup> s<sup>-1</sup> × 10<sup>6</sup>): (a) pre-satellite period (1979–1986); (b) SSM/I period (1988–1997); (c) AMSU period (1999–2010); (d) zonal average of ocean-only data with 12-month moving average filter applied, vertical lines indicate start of SSM/I data in August 1987 and AMSU in November 1998. The thick line in all panels shows the zero contour.

Relative to OAFlux zonal decadal mean,  $q_2$  from 20CRv2 is dry in tropical regions in every decade from the 1960s to the 2000s. Agreement between 20CRv2 and OAFlux is closer outside the Tropics but systematic differences are evident and remain similar in character throughout the five decades. For the 1980s comparison 20CRv2, MERRA and ERAI are all available and all show lower  $q_2$  than OAFlux in the Tropics. MERRA shows higher  $q_2$  in mid-latitudes and ERAI  $q_2$  is lower than OAFlux in the decadal zonal mean at all latitudes and all available decades. Agreement in zonal  $q_2$  and its temporal evolution among the datasets is weak. The increase in MERRA  $q_2$  relative to OAFlux, ERAI and 20CRv2 in the tropics in each decade is striking. This is consistent with the analysis increments for columnintegrated water vapour (Figure 6) which show that the assimilation of observations acts to increase the water vapour content of the MERRA atmosphere.

Figure 9 shows changes in decadal, zonal mean humidity for each product relative to its 1980s zonal mean. Plots in the first column present data masked as in Figures 7 and 8 (ice and land only), the plots in the other columns present data masked appropriately for the *in situ* products and are discussed in the next section. There is very little consistency among the reanalysis-based products as to how the zonal patterns of  $q_2$  change over time. 5.2. All products, masking for data uncertainty applied in addition to land/ice mask

Figure 10 shows the same information as Figure 7 but includes both *in situ* and reanalysis-based products and is masked as shown in Figure 1(d) and indicated in each row of Figure 10. Relatively high values for the FSUv3.0 product are seen in all regions, as noted by Smith *et al.* (2011). FSUv3.0 would be expected to contain higher  $q_{10}$  than NOCv2.0 as the adjustments applied to the NOCv2.0 dataset act, on average, to reduce  $q_{10}$ . However, the size of the adjustments (Figures 2 and 3) is smaller than differences between NOCv2.0 and FSUv3.0 (Figure 10). Differences between FSUv3.0 and the UWM/COADS dataset are also large in their period of overlap suggesting that the adjustments are not the cause of these differences.

The datasets shown in Figure 10 are a mixture of estimates of  $q_2$  and  $q_{10}$ . Although there is uncertainty in  $\Delta q_{2-10}$  (Figure 4), 2 m values are expected to be higher than 10 m values, and  $\Delta q_{2-10}$  should be largest in regions of high specific humidity. Figure 10 does not show a clear separation of products representing  $q_2$  and  $q_{10}$  indicating that uncertainties due to near-surface humidity gradients are only one contribution to the systematic differences among the datasets.

The changes seen in MERRA and ERAI bound those in the *in situ* datasets. In the Northern Extratropics (Figure 10(a)), the change in  $q_{10}$  over time in NOCv2.0 is more similar to that seen in OAFlux, 20CRv2 and MERRA than the smaller increase shown in ERAI.



Figure 7. Annual time series and seasonal cycles for reanalysis-based datasets. Note the different *y*-axis scaling. (a)  $25^{\circ}N-75^{\circ}N$ ; (b)  $25^{\circ}S-25^{\circ}N$ ; (c)  $75^{\circ}S-25^{\circ}S$ ; (d)  $75^{\circ}S-75^{\circ}N$ . OAFlux  $q_2$  (dark solid); ERAI  $q_2$  (mid dotted); 20CRv2  $q_2$  (light solid); MERRA  $q_2$  (light dotted); MERRA  $q_{10}$  (light dashed).

The difference between NOCv2.0 and UWM/COADS in the Northern Extratropics and Tropics (Figure 10(b)) in their period of overlap is consistent with the size of the bias adjustments applied in NOCv2.0 (Figure 2). In the Tropical average (Figure 10(b)), the increase in NOCv2.0  $q_{10}$  is similar to that seen in OAFlux and 20CRv2  $q_2$ . Compared with NOCv2.0  $q_{10}$  the increase seen in MERRA  $q_2$  is too large and that seen in ERAI is too small. There is only a small amount of data available in the Southern Extratropical comparison (Figure 10(c)) but as in the Northern Extratropics the changes seen in NOCv2.0  $q_{10}$  are more similar to those in OAFlux and 20CRv2 rather than MERRA or ERAI. The difference between NOCv2.0 and UWM/COADS is larger than the bias adjustments applied to NOCv2.0 and is likely the result of a combination of adjustments with additional data available in NOCv2.0 and the difference in dataset construction methods used. The global comparison (Figure 10(d)) is dominated by data

in the Northern Extratropics due to the *in situ* sampling pattern.

Figure 11 is similar to Figure 8 but for the in situ data products with regional masks applied (as shown in the last column of Figure 11). Again, OAFlux is used as a comparison reference and decadal zonal differences from OAFlux are plotted. Relative to OAFlux  $q_2$  in the 1960s UWM/COADS  $q_{10}$  is slightly drier in the Northern mid-latitudes, similar to 20CRv2. However, in the tropics UWM/COADS is higher than OAFlux whereas 20CRv2 is lower. The differences (relative to OAFlux) in the tropics would be further increased if  $\Delta q_{2-10}$  was taken into account (Figure 4). In the 1970s, NOC data become available and unadjusted, height-adjusted and fully adjusted versions of the NOCv2.0 dataset are plotted. The north-south variation seen in UWM/COADS relative to OAFlux is not seen in the unadjusted NOC values so is likely a result of increases in available data (COADS Release 1/1a vs ICOADS Release 2.4/2.5) or



Figure 8. Zonal mean differences from OAFlux for five decades 1960s–2000s, ice and land masks applied as shown in Figure 1(d). (a) OAFlux values 1960s–2000s; (b) 20CRv2 – OAFlux  $q_2$ , zonal mean for 1960s shown with and without filtering in latitude. Panels (c–f) show filtered values only; (c) product – OAFlux  $q_2$ ; zonal mean for 1970s; (d) as (c) for 1980s; (e) 1990s; (f) 2000s.

analysis method rather than adjustments applied to the observations in the NOC dataset. The main inputs to OAFlux in the 1970s are ERA-40 and NCEP1 that both make use of unadjusted humidity values and differences between OAFlux  $q_2$  and the fully-adjusted NOC  $q_{10}$ 

are small in the decadal zonal mean. In the 1980s, the FSU product becomes available and values are higher than the other *in situ*-based products as seen previously in Figure 10. Zonal differences from OAFlux are also much larger than for the other *in situ* datasets. The bias



Figure 9. Zonal mean  $q_2$  anomalies from 1980s average. Dark dotted, 1960s; solid, 1970s; dark dashed, 1990s; light solid, 2000s. First column (a, d, g, j) is for ice-free ocean, columns 2 and 3 are for combined *in situ* masked data as shown in Figure 1(d). Note that values are not plotted where the full decade is not available and HadCRUH contains only anomalies.

adjustments cause NOC-adjusted product to be lower than OAFlux in the tropics and diverges from the unadjusted NOCv2.0 and the UWM/COADS products. Similar features are seen in the 1990s and 2000s. Comparing Figures 8 and 11 (noting that the horizontal scales have the same range but different limits), it is clear that there is little consensus between the estimates of specific humidity from the different *in situ* and reanalysis-based products, their regional variations and long-term changes.

Figure 9 shows the decadal zonal changes in each reanalysis-based (column 2) and *in situ* dataset (column 3) relative to the dataset mean over the 1980s for the masked region shown in Figure 10(d). Also included in

Figure 9 is the change in the HadCRUH anomaly dataset between the 1980s and the 1990s. This is similar to that seen in both the NOCv2.0 and FSUv3.0 datasets. When the changes are compared relative to each datasets own zonal mean for the 1980s its hard to find a consistent picture of decadal changes. Changes in  $q_2$  between the 1960s and the 1980s in UWM/COADS are more similar to those in 20CRv2 than OAFlux. UWM/COADS and NOCv2.0 show similar change between the 1970s and the 1980s, which again are more similar to 20CRv2 than OAFlux. NOCv2.0, FSUv3.0 and HadCRUH show similar changes between the 1990s but the reanalysis-based datasets all show different changes





Figure 10. Annual time series and seasonal cycles for reanalysis-based and in situ datasets for masked regions shown. Note the different y-axis scaling. (a)  $25^{\circ}N-75^{\circ}N$ ; (b)  $25^{\circ}S-25^{\circ}N$ ; (c)  $75^{\circ}S-25^{\circ}S$ ; (d)  $75^{\circ}S-75^{\circ}N$ . OAFlux  $q_2$  (dark solid); ERAI  $q_2$  (dark dotted); 20CRv2  $q_2$  (light solid); MERRA  $q_2$  (light dotted); MERRA  $q_{10}$  (light dashed), NOCv2.0 q10 (mid dotted); FSUv3.0 q10 (mid dashed); UWM/COADS q10 (dark dashed).

and none are similar to the changes seen in the *in situ* datasets.

The *in situ*  $q_{10}$  estimates are all based on similar observations, although the volumes of data and how they are adjusted and combined to give monthly fields might be expected to give differences in the final estimates of order  $0.1 \text{ g kg}^{-1}$ . The increase in MERRA  $q_2$  of nearly  $0.8 \text{ g kg}^{-1}$  between the 1980s and the 2000s at the Equator is not seen in any of the other datasets and ERAI is the only dataset to show a decrease in  $q_2$  at the same time and place. The *in situ*-based datasets are fairly sparsely sampled in the Tropics but there are some well-sampled shipping lanes in Tropical regions (Berry and Kent, 2009). Large-scale decadal averages should therefore be broadly representative of conditions which suggests that the changes seen in MERRA and ERAI in the Tropics may not be realistic.

#### 6. Discussion and conclusions

A comparison of eight different datasets has revealed substantial structural uncertainty in marine near-surface humidity, a GCOS ECV. Traditionally, humidity observations have been made on VOS but their sampling has never been sufficient to produce global estimates of marine surface humidity to the accuracy required for surface flux estimation (WCRP, 1989; Berry and Kent, 2009). VOS have, however, provided observations of useful accuracy for decades in well-sampled regions (Berry and Kent, 2009). This is now at threat due to declining numbers of VOS and a lack of the observational metadata required to make adjustments to the humidity observations for height and observational bias (Kent et al., 2007; Berry and Kent, 2011). Increasing automation on VOS has led to a change in instrument types and the impact of these changes has not yet been assessed, partly because



Figure 11. Zonal mean differences from OAFlux for five decades 1960s-2000s, for in situ-based products masked as shown in Figure 10(d). (a) OAFlux values 1960s-2000s; (b)  $20CRv2 - OAFlux q_2$ , zonal mean for 1960s shown with and without filtering in latitude. Panels (c-f) show filtered values only; (c) product - OAFlux  $q_2$ ; zonal mean for 1970s; (d) as (c) for 1980s; (e) 1990s; (f) 2000s.

of the lack of observational metadata. Observations from moored buoys supplement the VOS observations from the 1990s but only limited assessments of their characteristics have been made (McPhaden *et al.*, 1998).

The FSU humidities are about  $0.8 \text{ g kg}^{-1}$  higher than the other *in situ* datasets; the reasons for this are currently

unknown. The differences between the NOCv2.0 and UWM/COADS *in situ* datasets are consistent with the size of the adjustments applied to the NOCv2.0 dataset and other differences due to input data and construction methods. The uncertainty estimates in the NOCv2.0 dataset in the best sampled regions are  $\sim 0.2 \text{ g kg}^{-1}$  which

is substantially smaller than the differences amongst the reanalysis-based datasets. There are also substantial differences amongst the datasets in terms of magnitude, regional mean values and changes over time.

The evolving observing system is known to impact both in situ and reanalysis datasets. ERAI and MERRA are both examples of state-of-the-art atmospheric reanalysis but show very different characteristics in terms of how marine surface-specific humidity changes over time. MERRA shows moistening over the period from 1979, especially in the Tropics, but ERAI indicates much smaller changes. The response of MERRA to the assimilation of observations can be examined using analysis increments and a change in the analysis increments for integrated water vapour in response to the availability of first SSMI and then AMSU-A has already been noted (Robertson et al., 2011). In contrast, observations from SSMI and AMSU-B have the most impact on the ERAI analysis (Andersson et al., 2007). There is an increasing divergence of q estimates from the two reanalyses over time at all latitudes. Trends are not quantified here because they are dominated by the model representation of El Niño-Southern Oscillation variability in q due to the short ( $\sim$ 30 year) period of record. The OAFlux blended analysis combines data from older reanalyses and satellite estimates of q with weights derived from comparison with in situ observations (Yu et al., 2008). OAFlux is closer in to the in situ datasets (UWM/COADS and NOCv2.0) than to the reanalyses despite being representative of  $q_2$  rather than  $q_{10}$ .

The in situ datasets UWM/COADS and NOCv2.0 show more similarity to OAFlux in magnitude and regional means than OAFlux does to the other reanalysisbased datasets. The zonal, decadal differences between OAFlux and 20CRv2 are similar from the 1960s to the 2000s suggesting that 20CRv2 shows a fairly consistent model bias over time relative to OAFlux. As noted earlier the response of MERRA and ERAI  $q_2$  to the increasing availability of satellite observations acts in opposite directions in Tropical regions. OAFlux is based on earlier versions of reanalyses which would not have been able to assimilate the new observations from AMSU. It is interesting that there appears to be better agreement between NOCv2.0 and OAFlux than with either MERRA or ERAI. OAFlux ingests ERA-40, NCEP1 and NCEP2 which are thought to be poorer estimates of surface humidity over land at least (Simmons et al., 2010). In the 2000s, MERRA zonal average  $q_2$  becomes more similar to OAFlux (and hence the in situ datasets) than in previous decades that may indicate that the assimilation of satellite observations has acted to some extent to correct model error in MERRA which would make trend estimation from MERRA inaccurate.

Although there is poor agreement amongst the different datasets as to the mean near-surface humidity, the representation of interannual variability and seasonal cycles over large areal averages is more consistent. This is because uncertainties in anomalies will be smaller than those in absolute values. Whilst there is uncertainty in the adjustments required for *in situ* observations due to the lack of observational metadata and the need to assess new sensor types the uncertainties are smaller than the differences seen between ERAI and MERRA. On the basis of the results presented here, the structural uncertainty (i.e. that due to choice of data, QC and gridding methodology) in the marine-specific humidity and its changes over time are large and unlikely to meet the accuracy requirements described in the introduction.

Comparison of humidity datasets is problematic because of the nonlinear relationship between the different variables that can be used to represent humidity and temperature, pressure and atmospheric stability. This means that any conversion between humidity variables, or adjustment to different reference heights, is only accurate when calculated at the native model resolution. It would facilitate long-term comparisons if monthly mean values of at least q and rh were made available by dataset providers at the standard marine reference level of 10 m in addition to the land reference level of 2 m above sea level. The need to convert between different reference levels adds uncertainty to the comparisons presented, but presently other differences between the datasets are larger.

Given the large differences amongst the different datasets it seems unlikely that any of the data sources will be reliable enough to provide humidity estimates accurate enough to allow the estimation of monthly mean fluxes to better than  $10 \text{ W m}^{-2}$ . This accuracy may, however, be achievable using *in situ* data for some regions and periods.

To improve the accuracy of the *in situ*  $q_{10}$  estimates it will be necessary to revisit the adjustments applied to the observations including an assessment of newer types of VOS humidity sensors. It will also require the development of an improved methodology to better handle missing metadata, including the exploration of dataset ensemble methods. An assessment of the accuracy of humidity measurements on moored buoys in the field would also provide valuable information. One resource that is so-far underutilised is the use of Research Vessel humidity measurements to provide an estimate of absolute biases in specific humidity.

## Acknowledgements

The International Comprehensive Ocean-atmosphere Data Set (ICOADS) archive of surface marine observations underpins the datasets used in this study. Scott Woodruff, Sandy Lubker, Steven Worley and Eric Freeman have all provided help and expert advice when needed. We are grateful to the National Center for Atmospheric Research for making the NOCv2.0 data available from the Computational and Information Systems Laboratory Research (http://rda.ucar.edu/datasets/ds260.3/). Data Archive UWM/COADS data were obtained from the International Research Institute for Climate and Society/Lamont-Doherty Earth Observatory Climate Data Library

(iridl.ldeo.columbia.edu/). HadCRUH data were downloaded from the Met Office Hadley Centre observations datasets webpages (http://www.metoffice.gov.uk/hadobs/ index.html). FSUv3.0 was downloaded from the Center for Atmosphere-ocean Prediction Studies of Florida State University (http://coaps.fsu.edu/RVSMDC/ FSUFluxes/index.php). Twentieth Century Reanalysis V2 data provided by the NOAA/OAR/ESRL PSD, Boulder, Colorado, USA, from their Web site at http://www.esrl. noaa.gov/psd/. OAFlux data were provided by the WHOI OAFlux project (http://oaflux.whoi.edu) funded by the NOAA Climate Observations and Monitoring (COM) programme. ERA Interim data were downloaded from the ECMWF data server. MERRA data were obtained from the Global Modeling and Assimilation Office (GMAO) and the NASA Goddard Earth Sciences (GES) Data and Information Services Center (DISC). We would like to thank the anonymous reviewers for their help in improving this article. This study received funding through the NERC National Centre for Earth Observation (NCEO) and the National Oceanography Centre's National Capability programme.

#### References

- Alexander RC, Mobley RL. 1976. Monthly average sea-surface temperatures and ice-pack limits on a 1° global grid. *Monthly Weather Review* **104**: 143–148. DOI: 10.1175/1520-0493(1976)104<0143:MASTAI>2.0.CO;2.
- Andersson E, Bauer P, Beljaars A, Chevallier F, Hólm E, Janisková M, Kållberg P, Kelly G, Lopez P, McNally A, Moreau E, Simmons AJ, Thépaut J-N, Tompkins AM. 2005. Assimilation and modeling of the atmospheric hydrological cycle in the ECMWF forecasting system. *Bulletin of the American Meteorological Society* 86(3): 387–402. DOI: 10.1175/BAMS-86-3-387.
- Andersson E, Hólm E, Bauer P, Beljaars A, Kelly GA, McNally AP, Simmons AJ, Thépaut J-N, Tompkins AM. 2007. Analysis and forecast impact of the main humidity observing systems. *Quarterly Journal of the Royal Meteorological Society* **133**: 1473–1485. DOI: 10.1002/qj.112.
- Andersson A, Fennig K, Klepp C, Bakan S, Grassl H, Schulz J. 2010. The Hamburg ocean atmosphere parameters and fluxes from satellite data - HOAPS-3. *Earth System Science Data Discussions* 3: 143–194. DOI: 10.5194/essdd-3-143-2010.
- Bentamy A, Katsaros KB, Alberto M, Drennan WM, Forde EB, Roquet H. 2003. Satellite estimates of wind speed and latent heat flux over the global oceans. *Journal of Climate* 16: 637–656.
- Berry DI, Kent EC. 2005. The effect of instrument exposure on marine air temperatures: an assessment using VOSClim data. *International Journal of Climatology* **25**(7): 1007–1022. DOI: 10.1002/joc.1178.
- Berry DI, Kent EC. 2009. A new air-sea interaction gridded dataset from ICOADS with uncertainty estimates. *Bulletin of the American Meteorological Society* **90**(5): 645–656. DOI: 10.1175/2008BAMS2639.1.
- Berry DI, Kent EC. 2011. Air-sea fluxes from ICOADS: the construction of a new gridded dataset with uncertainty estimates. *International Journal of Climatology* **31**(7): 987–1001. DOI: 10.1002/joc.2059.
- Berry DI, Kent EC, Taylor PK. 2004. An analytical model of heating errors in marine air temperatures from ships. *Journal* of Atmospheric and Oceanic Technology 21(8): 1198–1215. DOI: 10.1175%2 F1520-0426(2004)021%3C1198:AAMOHE% 3E2.0.CO;2.
- Bosilovich MG, Robertson FR, Chen J. 2011. Global energy and water budgets in MERRA. *Journal of Climate* 24: 5721–5739. DOI: 10.1175/2011JCLI4175.1.
- Bourassa MA, Romero R, Smith SR, O'Brien JJ. 2005. A new FSU winds climatology. *Journal of Climate* 18: 3686–3698. DOI: http://dx.doi.org/10.1175/JCLI3487.1.

- Bourassa MA, Gille ST, Jackson DL, Roberts JB, Wick GA. 2010. Ocean winds and turbulent air-sea fluxes inferred from remote sensing. *Oceanography* 23(4): 36–51. DOI: 10.5670/oceanog.2010.04.
- Bourlès B, Lumpkin R, McPhaden MJ, Hernandez F, Nobre P, Campos E, Yu L, Planton S, Busalacchi AJ, Moura AD, Servain J, Trotte J. 2008. The PIRATA program: history, accomplishments, and future directions. *Bulletin of the American Meteorological Society* 89: 1111–1125.
- Bradley EF, Fairall CW. 2007. A Guide to Making Climate Quality Meteorological and Flux Measurements at Sea. NOAA Technical Memorandum OAR PSD-311, NOAA/ESRL/PSD, Boulder, CO, 108.
- Breaker LC, Gilhousen DB, Burroughs LD. 1998. Preliminary results from long-term measurements of atmospheric moisture in the marine boundary layer in the Gulf of Mexico. *Journal of Atmospheric and Oceanic Technology* **15**: 661–676. DOI: http://dx.doi.org/10.1175/1520-0426(1998)015<0661:PRFLTM> 2.0.CO;2.
- Brunke MA, Wang Z, Zeng X, Bosilovich M, Shie C-L. 2011. An assessment of the uncertainties in ocean surface turbulent fluxes in 11 reanalysis, satellite-derived, and combined global datasets. *Journal of Climate* **24**: 5469–5493. DOI: 10.1175/2011JCLI4223.1.
- Chou S-H, Nelkin E, Ardizzone J, Atlas RM, Shie C-L. 2003. Surface turbulent heat and momentum fluxes over global oceans based on the goddard satellite retrievals, version 2 (GSSTF2). *Journal of Climate* **16**: 3256–3273. DOI: http://dx.doi.org/10.1175/1520-0442(2003)016<3256:STHAMF>2.0.CO;2.
- Compo GP, Whitaker JS, Sardeshmukh PD, Matsui N, Allan RJ, Yin X, Gleason BE, Vose RS, Rutledge G, Bessemoulin P, Brönnimann S, Brunet M, Crouthamel RI, Grant AN, Groisman PY, Jones PD, Kruk M, Kruger AC, Marshall GJ, Maugeri M, Mok HY, Nordli Ø, Ross TF, Trigo RM, Wang XL, Woodruff SD, Worley SJ. 2011. The twentieth century reanalysis project. *Quarterly Journal of the Royal Meteorological Society* **137**: 1–28. DOI: 10.1002/qj.776.
- Cressman GP. 1959. An operational objective analysis system. *Monthly Weather Review* **87**: 367–374. DOI: http://dx.doi.org/10.1175/1520-0493(1959)087<0367:AOOAS>2.0.CO;2.
- Da Silva AM, Young CC, Levitus S. 1994. *Algorithms and Procedures, Vol. 1, Atlas of Surface Marine Data 1994* NOAA Atlas NESDIS 6. US Government Printing Office: Washington DC; 83.
- Dai A. 2006. Recent climatology, variability and trends in global surface humidity. *Journal of Climate* 19: 3589–3606. DOI: 10.1175/JCLI3816.1.
- Dee D, National Center for Atmospheric Research Staff (eds). The Climate Data Guide: ERA-Interim. Retrieved September 25, 2012. Available at https://climatedataguide.ucar.edu/reanalysis/era-interim.
- Dee DP, Uppala SM, Simmons AJ, Berrisford P, Poli P, Kobayashi S, Andrae U, Balmaseda MA, Balsamo G, Bauer P, Bechtold P, Beljaars ACM, van de Berg L, Bidlot J, Bormann N, Delsol C, Dragani R, Fuentes M, Geer AJ, Haimberger L, Healy SB, Hersbach H, Hólm EV, Isaksen L, Kållberg P, Köhler M, Matricardi M, McNally AP, Monge-Sanz BM, Morcrette J-J, Park B-K, Peubey C, de Rosnay P, Tavolato C, Thépaut J-N, Vitart F. 2011. The ERA-Interim reanalysis: configuration and performance of the data assimilation system. *Quarterly Journal of the Royal Meteorological Society* 137: 553–597. DOI: 10.1002/qj.828.
- ECMWF. 2007. IFS DOCUMENTATION Cy31r1, Operational implementation 12 September 2006, Part IV, Physical processes, 155. Retrieved 21st May 2010. Available at http://www.ecmwf.int/research/ifsdocs/CY31r1/index.html.
- Fairall CW, Bradley EF, Rogers DP, Edson JB, Young GS. 1996. Bulk parameterization of air-sea fluxes for TOGA COARE. *Journal of Geophysical Research* 101: 3747–3764.
- Fairall CW, Bradley EF, Hare JE, Grachev AA, Edson JB. 2003. Bulk parameterization of air–sea fluxes: updates and verification for the COARE algorithm. *Journal of Climate* **16**: 571–591. DOI: 10.1175/1520-0442(2003)016<0571:BPOASF>2.0.CO;2.
- Fairall C, Barnier B, Berry DI, Bourassa MA, Bradley F, Clayson CA, de Leeuw G, Drennan WM, Gille ST, Gulev SK, Kent EC, McGillis WR, Quartly GD, Ryabinin V, Smith SR, Weller RA, Yelland MJ, Zhang H-M. 2010. Observations to quantify air-sea fluxes and their role in climate variability and predictability. In *Proceedings of OceanObs'09: Sustained Ocean Observations and Information for Society*, Vol. 2, Hall J, Harrison DE, Stammer D (eds). ESA Publication WPP-306: Venice, Italy.
- GCOS. 2010. Implementation Plan for the Global Observing System for Climate in Support of the UNFCCC, GCOS-138, WMO Geneva, 180. Retrieved November 21, 2012. Available at http://www.wmo.int/pages/prog/gcos/Publications/gcos-138.pdf.

- Henderson-Sellers B. 1984. A new formula for latent heat of vaporization of water as a function of temperature. *Quarterly Journal of the Royal Meteorological Society* **111**: 1186–1190. DOI: 10.1002/qj.49711046626.
- Josey SA, Kent EC, Taylor PK. 1999. New insights into the ocean heat budget closure problem from analysis of the SOC air-sea flux climatology. *Journal of Climate* **12**(9): 2856–2880. DOI: 10.1175/1520-0442(1999)012<2856:NIITOH>2.0.CO;2.
- Kalnay E, Kanamitsu M, Kistler R, Collins W, Deaven D, Gandin L, Iredell M, Saha S, White G, Woolen J, Zhu Y, Chelliah M, Ebisuzaki W, Higgins W, Janowiak J, Mo KC, Ropelewski C, Wang J, Leetma A, Reynolds RW, Jenne R, Joseph D. 1996. The NCEP/NCAR 40-year reanalysis project. *Bulletin of the American Meteorological Society* **77**: 437–471. DOI: 1520-0477(1996)077<0437:TNYRP>2.0.CO;2.
- Kanamitsu M, Ebisuzaki W, Woollen J, Yang S-K, Hnilo JJ, Fiorino M, Potter GL. 2002. NCEP–DOE AMIP-II reanalysis (R-2). *Bulletin of the American Meteorological Society* 83: 1631–1643. DOI: 10.1175/BAMS-83-11-1631.
- Kennedy JJ, Thorne PW, Peterson TC, Ruedy RA, Stott PA, Parker DE, Good SA, Titchner HA, Willett KM. 2010. How Do We Know the World Has Warmed?, in Arndt, D. S., M. O. Baringer, and M. R. Johnson, Eds., 2010: State of the climate in 2009. *Bulletin of the American Meteorological Society* **91**(7): S26.
- Kennedy JJ, Rayner NA, Smith RO, Saunby M, Parker DE. 2011. Reassessing biases and other uncertainties in sea-surface temperature observations since 1850 part 2: biases and homogenisation. *Journal of Geophysical Research* **116**: D14104. DOI: 10.1029/2010JD015220.
- Kent EC, Berry DI. 2005. Quantifying random errors in voluntary observing ships meteorological observations. *International Journal* of Climatology 25(7): 843–856. DOI: 10.1002/joc.1167.
- Kent EC, Ingleby B. 2010. From observations to forecasts part 6. Marine meteorological observations. *Weather* 65(9): 231–238. DOI: 10.1002/wea.625.
- Kent EC, Taylor PK. 1996. Accuracy of humidity measurements on ships: Consideration of solar radiation effects. *Journal of Atmospheric and Oceanic Technology* **13**(6): 1317–1321. DOI: 10.1175/1520-0426(1996)013<1317:AOHMOS>2.0.CO;2.
- Kent EC, Taylor PK, Truscott BS, Hopkins JS. 1993a. The accuracy of voluntary observing ship's meteorological observations - Results of the VSOP-NA. *Journal of Atmospheric* and Oceanic Technology **10**(4): 591–608. DOI: 10.1175/1520-0426(1993)010<0591:TAOVOS>2.0.CO;2.
- Kent EC, Tiddy RJ, Taylor PK. 1993b. Correction of marine air temperature observations for solar radiation effects. *Journal of Atmospheric* and Oceanic Technology **10**(6): 900–906. DOI: 10.1175/1520-0426(1993)010<0900:COMATO>2.0.CO;2.
- Kent EC, Taylor PK, Challenor PG. 2000. The effect of successive correction on variability estimates for climatological datasets. *Journal of Climate* 13(11): 1845–1857. DOI: 10.1175/1520-0442(2000)013<1845:TEOSCO>2.0.CO;2.
- Kent EC, Woodruff SD, Berry DI. 2007. WMO Publication No. 47 metadata and an assessment of voluntary observing ships observation heights in ICOADS. *Journal of Atmospheric and Oceanic Technology* 24(2): 214–234. DOI: 10.1175/JTECH1949.1.
- Kubota M, Iwasaka N, Kizu S, Konda M, Kutsuwada K. 2002. Japanese ocean flux datasets with use of remote sensing observations (J-OFURO). *Journal of Oceanography* 58: 213–225.
- Maury MF. 1851. Explanations and Sailing Directions to Accompany the Wind and Current Charts, 3rd edn. C. Alexander, Printer: Washington; 315 + plates.
- McPhaden MJ, Busalacchi AJ, Cheney R, Donguy J-R, Gage KS, Halpern D, Ji M, Julian P, Meyers G, Mitchum GT, Niiler PP, Picaut J, Reynolds RW, Smith N, Takeuchi K. 1998. The tropical oceanglobal atmosphere (TOGA) observing system: a decade of progress. *Journal of Geophysical Research* 103: 14,169–14,240.
  McPhaden MJ, Meyers G, Ando K, Masumoto Y, Murty VSN, WCPhaden MJ, Meyers G, Ando K, Masumoto Y, Murty VSN,
- McPhaden MJ, Meyers G, Ando K, Masumoto Y, Murty VSN, Ravichandran M, Syamsudin F, Vialard J, Yu L, Yu W. 2009. RAMA: the research moored array for African-Asian-Australian monsoon analysis and prediction. *Bulletin of the American Meteorological Society* **90**: 459–480.
- Met Office. 1995. Marine observer's guide, 11th edn. HMSO: London; -236.
- Rayner NA, Parker DE, Horton EB, Folland CK, Alexander LV, Rowell DP, Kent EC, Kaplan A. 2003. Global analyses of SST, sea ice and night marine air temperature since the late 19th century. *Journal of Geophysical Research* **108**(D14): 4407. DOI: 10.1029/2002JD002670.

- Rayner NA, Brohan P, Parker DE, Folland CK, Kennedy JJ, Vanicek M, Ansell T, Tett SFB. 2006. Improved analyses of changes and uncertainties in sea surface temperature measured *in situ* since the mid-nineteenth century: the HadSST2 data set. *Journal of Climate* 19(3): 446–469. DOI: 10.1175/JCLI3637.1.
- Reynolds RW, Rayner NA, Smith TM, Stokes DC, Wang W. 2002. An improved *in situ* and satellite SST analysis for climate. *Journal of Climate* **15**: 1609–1625.
- Rienecker MM, Suarez MJ, Todling R, Bacmeister J, Takacs L, Liu H-C, Gu W, Sienkiewicz M, Koster RD, Gelaro R, Stajner I, Nielsen JE. 2008. The GEOS-5 Data Assimilation System Documentation of versions 5.0.1 and 5.1.0, and 5.2.0. NASA Tech. Rep. Series on Global Modeling and Data Assimilation, NASA/TM-2008-104606: 27: 92. Retrieved November 21, 2010. Available at gmao.gsfc.nasa.gov/pubs/docs/GEOS5\_104606-Vol27.pdf.
- Rienecker MM, Suarez MJ, Gelaro R, Todling R, Bacmeister J, Liu E, Bosilovich MG, Schubert SD, Takacs L, Kim G-K, Bloom S, Chen J, Collins D, Conaty A, da Silva A, Gu W, Joiner J, Koster RD, Lucchesi R, Molod A, Owens T, Pawson S, Pegion P, Redder CR, Reichle R, Robertson FR, Ruddick AG, Sienkiewicz M, Woollen J. 2011. MERRA: NASA's modern-era retrospective analysis for research and applications. *Journal of Climate* 24: 3624–3648. DOI: 10.1175/JCLI-D-11-00015.1.
- Roberts JB, Clayson CA, Robertson FR, Jackson DL. 2010. Predicting near-surface atmospheric variables from special sensor microwave/imager using neural networks with a first-guess approach. *Journal of Geophysical Research-Atmospheres* **115**: D19113. DOI: 10.1029/2009JD013099.
- Roberts JB, Robertson FR, Clayson CA, Bosilovich MG. 2012. Characterization of turbulent latent and sensible heat flux exchange between the atmosphere and ocean in MERRA. *Journal of Climate* 25: 821–838. DOI: 10.1175/JCLI-D-11-00029.1.
- Robertson FR, Bosilovich MG, Chen J, Miller TL. 2011. The effect of satellite observing system changes on MERRA water and energy fluxes. *Journal of Climate* 24: 5197–5217.
- Shie C-L, Chiu LS, Adler R, Gao S, Chokngamwong R, Lin I-I, Nelkin E, Ardizzone J, Xie P, Wang F-C. 2010 A recently revived and produced global air-sea surface turbulent fluxes dataset – GSSTF2b: Validations and Findings., Proceedings of the Joint 2010 CWB Weather Analysis and Forecasting & COAA 5th International Ocean–atmosphere Conference, 307–312, Taipei, Taiwan, June 28–30, 2010. Retrieved November 21, 2012. Available at http://disc.sci.gsfc.nasa.gov/measures/documentation/Shie-etal-COAA2010.pdf.
- Simmons AJ, Willett KM, Jones PD, Thorne PW, Dee DP. 2010. Lowfrequency variations in surface atmospheric humidity, temperature, and precipitation: Inferences from reanalyses and monthly gridded observational data sets. *Journal of Geophysical Research* 115: D01110. DOI: 10.1029/2009JD012442.
- Smith SD. 1980. Wind stress and heat flux over the ocean in gale force winds. *Journal of Physical Oceanography* **10**: 709–726. DOI: 10.1175/1520-0485(1980)010<0709:WSAHFO>2.0.CO;2.
- Smith SD. 1988. Coefficients for sea surface wind stress, heat flux, and wind profiles as a function of wind speed and temperature. *Journal of Geophysical Research* **93**(C12): 15467–15472. DOI: 10.1029/JC093iC12p15467.
- Smith SR, Hughes PJ, Bourassa MA. 2011. A comparison of nine monthly air-sea flux products. *International Journal of Climatology* 31(7): 1002–1027. DOI: 10.1002/joc.2225.
- Sterl A. 2004. On the (in)homogeneity of reanalysis products. *Journal of Climate* 17: 3866–3873. DOI: http://dx.doi.org/10.1175/1520-0442(2004)017<3866:OTIORP>2.0.CO;2.
- Trenberth KE, Smith L, Qian T, Dai A, Fasullo J. 2007. Estimates of the global water budget and its annual cycle using observational and model data. *Journal of Hydrometeorology* 8: 758–769. DOI: 10.1175/JHM600.1.
- Trenberth K, Dole R, Xue Y, Onogi K, Dee D, Balmaseda M, Bosilovich M, Schubert S, Large L. 2010. Atmospheric reanalyses: a major resource for ocean product development and modeling. In *Proceedings of OceanObs'09: Sustained Ocean Observations and Information for Society*, Vol. 2, Hall J, Harrison DE, Stammer D (eds). ESA Publication WPP-306: Venice, Italy. DOI: 10.5270/OceanObs09.cwp.90.
- Uppala SM, Kållberg PW, Simmons AJ, Andrae U, da Costa Bechtold V, Fiorino M, Gibson JK, Haseler J, Hernandez A, Kelly GA, Li X, Onogi K, Saarinen S, Sokka N, Allan RP, Andersson E, Arpe K, Balmaseda MA, Beljaars ACM, van de Berg L, Bidlot J, Bormann N, Caires S, Chevallier F, Dethof A, Dragosavac M, Fisher M, Fuentes

M, Hagemann S, Hólm E, Hoskins BJ, Isaksen L, Janssen PAEM, Jenne R, McNally AP, Mahfouf J-F, Morcrette J-J, Rayner NA, Saunders RW, Simon P, Sterl A, Trenberth KE, Untch A, Vasiljevic D, Viterbo P, Woollen J. 2005. The ERA-40 reanalysis. *Quarterly Journal of the Royal Meteorological Society* **131**: 2961–3012. DOI: 10.1256/qj.04.176.

- Wang W, Xie P, Yoo S-H, Xue Y, Kumar A, Wu X. 2010. An assessment of the surface climate in the NCEP climate forecast system reanalysis. *Climate Dynamics* 37: 1601–1620. DOI: 10.1007/s00382-010-0935-7.
- WCRP. 1989. WOCE Surface Flux Determinations A strategy for in situ measurements. Working Group on in situ measurements for Fluxes WCRP-23 (WMO/TD No.304), WMO, Geneva.
- Webster PA, Lukas R. 1992. The coupled ocean-atmosphere response experiment. Bulletin of the American Meteorological Society 73: 1377–1416.
- Weller RA, Bradley EF, Lukas R. 2004. The interface or air-sea flux component of the TOGA coupled ocean–atmosphere response experiment and its impact on subsequent air-sea interaction studies. *Journal of Atmospheric and Oceanic Technology* **21**: 223–257.
- Weller RA, Bradley EF, Edson JB, Fairall CW, Brooks I, Yelland MJ, Pascal RW. 2008. Sensors for physical fluxes at the sea surface: energy, heat, water, salt. *Ocean Science* 4(4): 247–263. DOI: 10.5194/os-4-247-2008.
- WGASF. 2000. Intercomparison and validation of ocean-atmosphere energy flux fields – Final report of the Joint WCRP/SCOR Working Group on Air-Sea Fluxes. WCRP-112, WMO/TD-1036. Taylor PK (ed), 306. Retrieved November 21, 2012. Available at http://eprints.soton.ac.uk/69522/1/wgasf\_final\_rep.pdf.
- Willett KM, Gillett NP, Jones PD, Thorne PW. 2007. Attribution of observed surface humidity changes to human influence. *Nature* 449: 710–712. DOI: 10.1038/nature06207.
- Willett KM, Jones PD, Gillett NP, Thorne PW. 2008. Recent changes in surface humidity: development of the HadCRUH dataset. *Journal* of Climate 21(20): 5364–5383. DOI: 10.1175/2008JCLI2274.1.

- Willett KM, Berry DI, Simmons A. 2012. Global climate; hydrological cycle surface humidity in "State of the Climate in 2011". Bulletin of the American Meteorological Society 93(7): S23–S25.
- WMO. 2008. Guide to Meteorological Instruments and Methods of Observation, Part I, Measurement of Meteorological Variables, Chapter 4 Measurement of Humidity, 7th edition, World Meteorological Organization I.4: 1–30. Retrieved November 21, 2012. Available at http://www.wmo.int/pages/prog/www/IMOP/CIMO-Guide.html.
- Woodruff SD, Slutz RJ, Jenne RL, Steurer PM. 1987. A comprehensive ocean–atmosphere data set. *Bulletin of the American Meteorological Society* 68: 1239–1250.
- Woodruff SD, Lubker SJ, Wolter K, Worley SJ, Elms JD. 1993. Comprehensive ocean-atmosphere data set (COADS) release 1a: 1980–92. *Earth System Monitor* **4**(1): 1–8.
- Woodruff SD, Worley SJ, Lubker SJ, Ji Z, Freeman JE, Berry DI, Brohan P, Kent EC, Reynolds RW, Smith SR, Wilkinson C. 2011. ICOADS release 2.5 and data characteristics. *International Journal* of Climatology **31**(7): 951–967. DOI: 10.1002/joc.2103.
- Worley SJ, Woodruff SD, Reynolds RW, Lubker SJ, Lott N. 2005. ICOADS release 2.1 data and products. *International Journal of Climatology* 25: 823–842.
- Yu L, Weller RA. 2007. Objectively analyzed air-sea heat fluxes for the global ice-free oceans (1981–2005). Bulletin of the American Meteorological Society 88: 527–539.
- Yu L, Jin X, Weller RA. 2008. Multidecade Global Flux Datasets from the Objectively Analyzed Air-sea Fluxes (OAFlux) Project: Latent and sensible heat fluxes, ocean evaporation, and related surface meteorological variables. Woods Hole Oceanographic Institution, OAFlux Project Technical Report. OA-2008-01, 64. Woods Hole. Massachusetts. Retrieved November21, 2012. Available at http://oaflux.whoi.edu/pdfs/OAFlux\_TechReport\_3rd\_release.pdf.